OPEN

Modeling the Cost-Effectiveness of Home-Based HIV Testing and Education (HOPE) for Pregnant Women and Their Male Partners in Nyanza Province, Kenya

Monisha Sharma, ScM,* Carey Farquhar, MD, MPH,*[†]‡ Roger Ying, BS,[†] Daisy Krakowiak, MPH, PhD,* John Kinuthia, MBChB, MMed, MPH,[†]§||¶ Alfred Osoti, MBChB, MMed, MPH,*||¶ Victor Asila, BSc,¶ Molly Gone, BSc,§ Jennifer Mark, MPH,* and Ruanne V. Barnabas, MD, DPhil*[†]‡#

Introduction: Women in sub-Saharan Africa face a 2-fold higher risk of HIV acquisition during pregnancy and postpartum and the majority do not know the HIV status of their male partner. Homebased couple HIV testing for pregnant women can reduce HIV transmission to women and infants while increasing antiretroviral therapy (ART) coverage in men. However, the cost-effectiveness of this program has not been evaluated.

Methods: We modeled the health and economic impact of implementing a home-based partner education and HIV testing (HOPE) intervention for pregnant women and their male partners in a region of

From the *Department of Epidemiology, University of Washington, Seattle, WA; †Department of Global Health, University of Washington, Seattle, WA; ‡School of Medicine, University of Washington, Seattle, WA; Departments of §Research and Programs; ||Reproductive Health, Kenyatta National Hospital, Nairobi, Kenya; ¶Department of Obstetrics and Gynecology, University of Nairobi, Nairobi, Kenya; and #Vaccine and Infectious Diseases Division, Fred Hutchinson Cancer Research Center, Seattle, WA.

Supported by the U.S. National Institutes of Health (NIH) Grant R01HD075108. MS received support from the Center for AIDS and STD Training Grant (NIAID T32A107140). CF received support from NIH Grant K24A1087399. D.K. received support from Achievement Rewards for College Scientists (ARCS) Fellowship, the University of Washington Top Scholar Award and Global Opportunities Health Fellowship (GO Health). The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the article.

Related presentations: Sharma M, Farquhar C, Kinuthia J, Osoti A, Asila V, Parikh S, Krakowiak D, Gone A, Barnabas R.V. Cost of home HIV testing and education for male partners of pregnant women in Kenya. [Poster] HIV Pathogenesis, Treatment and Prevention (IAS), Vancouver, BC, July 20, 2015. The authors have no funding or conflicts of interest to disclose.

Supplemental digital content is available for this article. Direct URL citations appear in the printed text and are provided in the HTML and PDF

versions of this article on the journal's Web site (www.jaids.com). C.F. and R.B. conceived of the analysis. M.S. conducted the microcosting and costeffectiveness analyses. R.Y. and M.S. conducted the modeling. MS wrote the manuscript with input from C.F., R.Y., D.K., J.K., A.O., V.A., M.G., J.M., and R.B.

This study was approved by the Kenya Ethics Review Committee and the University of Washington Human Subjects Division.

Clinicaltrials.gov registry: NCT01784783.

Correspondence to: Monisha Sharma, ScM Department of Epidemiology, University of Washington, 1959 NE Pacific Street, Seattle, WA 98195 (e-mail: msharma04@gmail.com).

Copyright © 2016 Wolters Kluwer Health, Inc. All rights reserved. This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivatives License 4.0 (CC BY-NC-ND), which permits downloading and sharing the work provided it is properly cited. The work cannot be changed in any way or used commercially.

Western Kenya (formally Nyanza Province). We used data from the HOPE randomized clinical trial conducted in Kisumu, Kenya, to parameterize a mathematical model of HIV transmission. We conducted an in-country microcosting of the HOPE intervention (payer perspective) to estimate program costs as well as a lower cost scenario of task-shifting to community health workers.

Results: The incremental cost of adding the HOPE intervention to standard antenatal care was \$31–37 and \$14–16 USD per couple tested with program and task-shifting costs, respectively. At 60% coverage of male partners, HOPE was projected to avert 6987 HIV infections and 2603 deaths in Nyanza province over 10 years with an incremental cost-effectiveness ratio (ICER) of \$886 and \$615 per disability-adjusted life year averted for the program and task-shifting scenario, respectively. ICERs were robust to changes in intervention coverage, effectiveness, and ART initiation and dropout rates.

Conclusions: The HOPE intervention can moderately decrease HIV-associated morbidity and mortality by increasing ART coverage in male partners of pregnant women. ICERs fall below Kenya's per capita gross domestic product (\$1358) and are therefore considered cost-effective. Task-shifting to community health workers can increase intervention affordability and feasibility.

Key Words: HIV counseling and testing, sub-Saharan Africa, costeffectiveness, males, pregnant, PMTCT

(J Acquir Immune Defic Syndr 2016;72:S174–S180)

INTRODUCTION

HIV is one of the most serious health and economic challenges in sub-Saharan Africa (SSA), where over 70% of the world's new HIV infections occur.¹ Pregnant and post-partum women in SSA have particularly high rates of HIV acquisition; cohort studies report HIV incidences ranging from 2.3 to 7.6 per 100 person-years during pregnancy and post-partum,^{2–5} more than 2-fold higher than in nonpregnant women.⁶ This increased risk persists after adjustment for sexual behavior, which suggests that biological factors including hormonal or immunological changes during pregnancy may play a role in heightening susceptibility.⁶ With fertility rates around 5 children per woman in SSA, this can translate into 10 or more years of increased risk. HIV-infected women in East Africa also face more than 7-fold higher risk of maternal mortality compared with uninfected pregnant women,⁷ and

approximately 17% of HIV-infected mothers in SSA transmit HIV to their infants.⁸

Although pregnant women have high rates of HIV testing coverage in SSA largely due to antenatal testing, they are less likely to know the HIV status of their male partners. For example, in Kenya, during 2013, 88% of pregnant women were tested for HIV, whereas only 4.5% of their male partners underwent testing in the previous 12 months.^{8,9} This is a significant problem in generalized epidemic settings, because persons are most likely to become infected in stable partnerships. A modeling analysis using sexual behavior survey data from 18 countries in SSA suggests that HIV transmission within couples is largely transmitted from men to women, and men are more likely than women to become infected by someone other than their main partner, both of which indicate that interventions to reduce HIV in women will likely need to target men as well.¹⁰

As facility-based HIV testing and counseling (HTC) has not achieved high testing coverage in male partners of pregnant women in SSA, novel testing interventions are needed.^{11,12} Community-based HTC is a promising strategy that achieves high uptake and identifies asymptomatic HIV-positive individuals at higher CD4 counts than facility HTC.^{13,14} One form of community HTC, home-based HIV couple counseling for pregnant women and their male partners, can attain high HIV testing coverage in both pregnant women and their partners. Couple HIV testing and disclosure has been shown to increase women's adherence to both antiretroviral therapy (ART) and prevention of mother to child transmission regimens, including Option B+ (lifelong ART for HIV-infected pregnant women).¹⁵⁻

Increasing HIV testing in males reduces transmission to their female partners, while also averting morbidity and mortality in men associated with late ART initiation. Men are less likely than women to undergo HIV testing, less likely to start ART, and more likely to seek care at advanced disease stages, interrupt treatment, and die on ART.¹⁸ Antenatal clinic attendance is high (95.4% of pregnant women in Kenya, 2012), indicating the potential to achieve high coverage of male partners.¹⁹ However, home-based couple counseling for pregnant women is resource intensive, and its cost-effectiveness (ie, value for money) has not been well evaluated. We used a mathematical model to assess the cost-effectiveness of providing home-based partner education and HIV testing (HOPE) to couples as a part of routine antenatal care in Western Kenya. Primary cost and effectiveness data were collected from the HOPE study, a randomized controlled trial conducted in Kisumu, Kenya, in which couples either received the HOPE intervention (HOPE arm) or written invitations for male partners to attend clinic (INVITE arm). In addition to HIV education and testing, couples in the HOPE arm received information on facility delivery, exclusive breastfeeding, family planning, and voluntary medical male circumcision. In this analysis, we project the health and economic impact of implementing the HOPE intervention in the former Nyanza Province, a region in Western Kenya with high HIV prevalence (15.1%). From an implementation science perspective, we sought to translate the HOPE clinical trial results into projections of HIV burden averted and costs incurred under realistic program scale-up. Our analysis can be useful to policy makers charged with implementing evidence-based HIV interventions that maximize health benefits within a fixed budget.

METHODS

Home-Based Partner Education and Testing (HOPE) Intervention

Costs and effectiveness data were obtained from the HOPE intervention, a randomized controlled trial conducted from September 2013 to June 2015 in Kisumu, Kenya. Study design and outcomes are described in detail in this supplement.²⁰ Briefly, 601 pregnant women in stable partnerships were enrolled when they presented for antenatal care at Kisumu County Hospital and randomized to HOPE (intervention) or INVITE (control) arm. Couples in the intervention arm received a home visit in which study staff (health advisors) screened male partners and offered couples HIV counseling and testing. Health advisors also administered a standardized health education intervention on the importance of facility delivery, exclusive breastfeeding, family planning, and methods to prevent HIV transmission. Women in the INVITE arm were asked through written invitation to bring their partners to the clinic for couple HIV testing. The study compared uptake of male partner HIV testing, condom use, facility delivery, exclusive breastfeeding, HIV transmission to infants and mothers, and contraceptive use in the 2 arms. Outcomes were assessed at 6 weeks and again at 6 months after delivery. The HOPE intervention was found to increase male partner testing by more than 2-fold (relative risk: 2.1, 95% confidence interval: 1.81 to 2.42) and identified more HIV serodiscordant couples (13%) compared with the INVITE arm (4%). The HOPE intervention was not associated with an increase in intimate partner violence or other adverse events.20

Mathematical Model

We adapted a previously published dynamic HIV transmission model to include pregnancy states.²¹ The model was parameterized with epidemiologic data from the former Nyanza Province and calibrated to fit HIV incidence and prevalence from that region. The model simulates the natural history of HIV/AIDS using 3-month time steps. Men and women are stratified by 5-year age group (0-59 years), sexual activity (low, medium, and high), circumcision status, viral load (6 stages), and CD4 count (6 stages). Sexual behavior is assumed to change over time as individuals age. Susceptible individuals can acquire HIV and transition to acute infection. CD4 count declines over time based on clinical estimates from a prospective cohort.²¹ Persons on ART have a 96% reduction in transmission risk.²² Disability weights are assigned to each HIV state.²³ In addition to background mortality, HIV-positive individuals face a disease-specific mortality that varies by age and CD4 count.²⁴ Persons on ART are assumed to have the same mortality rates as those who are uninfected.²⁵ Dropout of ART occurs yearly and individuals are assumed to return to the CD4 count and viral load status they had before initiating ART. The model estimates the force of HIV infection as a function of sexual mixing (by age and sexual activity), proportion of HIVinfected individuals, circumcision, and HIV transmission probability. HIV-positive women who are not on ART have a probability of transmitting HIV to their infants. Changes in the population over time are estimated using a system of ordinary differential equations that are solved in MATLAB version 2015a using 4th-order Runge–Kutta methods.²⁶ Before projecting the impact of the HOPE intervention, the model was calibrated to reflect the age-specific and overall HIV prevalence from Nyanza province and CD4 distribution and ART coverage from Kenya. Additional details about the model, parameters, and calibration results are available in the Supplemental Digital Content, http://links.lww.com/QAI/A828.

Status Quo and Intervention Scenarios

For the status quo (no intervention) scenario, we modeled the impact of continued facility HIV testing and ART expansion at current scale-up rates.²⁷ In the intervention scenario, HOPE is added to the status quo scenario with 60% coverage of male partners of pregnant women, based on coverage from the trial. However, as coverage could either be higher or lower if implemented as a government program, we conducted sensitivity analyses varying coverage from 40% to 80%. We assumed that the HOPE intervention increased the ART initiation rate for HIV-positive male partners by 2.1-fold (the increase in male partner HIV testing found in the clinical trial).²⁰

Microcosting

A detailed microcosting was conducted following established guidelines for costing HIV interventions.^{28,29} Primary cost data were collected from budgets, expense reports staff, and local expert interviews. Costs were divided into mutually exclusive categories of: personnel, transportation, equipment, supplies, buildings and overhead, start-up, and phones and data monitoring. Time and motion studies were conducted over 3 weeks (June 10-30, 2014) to record staff time spent on intervention activities (eg, conducting HTC, tracing male partners, traveling to couples' homes). Research time (administering informed consent, reimbursements, etc) and other research cost were removed from the programmatic costs. The time and motion studies and interviews with staff were used to inform efficiency assumptions about the mean number of couples that could be tested per day. Capital costs, software development, and start-up cost (staff hiring, training, and community mobilization) were annualized assuming a 5-year useful life expectancy discounted annually at 3%. Costs were inflated to 2014 US dollars (USD) using the Kenya consumer price index. Total program costs were divided by the number of couples tested by HIV status under each scenario to determine the cost per person tested. Other costs including facility HIV testing, ART, and HIV/AIDS-related hospitalizations were estimated from the literature³⁰⁻³²; additional information is available in the Supplemental Digital Content, http://links.lww.com/QAI/A828.

Cost-Effectiveness Analysis

We calculated the incremental cost-effectiveness ratio (ICER) for adding the HOPE intervention to standard

 $S176 \mid$ www.jaids.com

antenatal care for disability-adjusted life years (DALYs) averted over 10 years. The ICER is measured as the additional cost divided by the additional health benefit of the intervention strategy compared with the next less costly strategy (the status quo). Consistent with health economic conventions, we considered the intervention to be very cost-effective if the ICER is less than Kenya's 2014 GDP per capita (1358 USD).³³ We used a 10-year time horizon as is common in cost-effectiveness analyses of HIV prevention.^{34,35}

Sensitivity Analyses

To explore the impact of key assumptions on our findings, we conducted sensitivity analyses varying coverage, costs, intervention effectiveness, projected ART expansion, and task-shifting from health advisors to community health workers. We also evaluated a scenario in which the HOPE intervention increased adherence to Option B+ as studies found that couple HIV testing and disclosure increases women's adherence.¹⁵

RESULTS

Microcosting

Costs were estimated for 2 scenarios: (1) a higher cost program model, reflecting the staff cadre of the research study -nurses enrolling pregnant women into the intervention, highly trained health advisors administering the intervention, and higher cost supplies including research-compatible mobile phones, and (2) a lower-cost task-shifting model in which nurses and health advisors are replaced with community health workers, the field coordinator is replaced with a community health worker manager, the data manager is reduced to a halftime position, and lower cost supplies are used (Table 1). Results of the time and motion observations showed that the HOPE educational component and couple HTC together take approximately 1 hour per couple (15-30 minutes longer for couples with discordant HIV status who require additional counseling). After accounting for travel time to participant's home, follow-ups, paperwork, and other staff responsibilities, we estimated that health advisors could test 3 couples per day. Staff were assumed to work 7 hours per day for 215 days per year (after accounting for holidays, vacation, and sick time). Supply costs per person tested included gloves, HIV screening test kit, and alcohol swabs. Additional supplies for HIV-positive persons included confirmatory test, and tie breaker test (assumed to be used in 5% of all HIVpositive cases). Supply wastage was assumed to be 5%. Transport costs included motorcycles used to travel to participants homes to conduct the intervention. Economic costs were estimated for donated goods, including hospital space. Incremental costs per couple tested were estimated separately for HIV discordant and concordant couples as the former required more staff time for counseling. Costs per couple tested ranged from \$31-37 for the program model and \$14-16 for the task-shifting model. Staff salaries represented the bulk of the costs (65%–70%) (Table 1).

Copyright © 2016 Wolters Kluwer Health, Inc. All rights reserved.

		Program Model		Task-Shifting Model				
	Concordant HIV-	Concordant HIV+	Discordant	Concordant HIV-	Concordant HIV+	Discordant		
Personnel	22.90	22.90	25.45	9.27	9.27	10.30		
Transportation	2.16	2.16	2.40	2.16	2.16	2.40		
Equipment	0.17	0.17	0.18	0.17	0.17	0.19		
Supplies	3.63	7.65	6.27	0.53	0.89	0.79		
Buildings and overhead	0.96	0.96	1.07	0.96	0.96	1.07		
Start-up	0.44	0.44	0.49	0.27	0.27	0.30		
Data capture and use	0.52	0.52	0.57	0.44	0.44	0.49		
Total (per couple tested)	30.78	34.80	36.43	13.81	14.17	15.55		

Model Estimated Health and Economic Impact of the HOPE Intervention

In the base-case scenario, assuming the HOPE intervention increased male partner ART initiation by 2.1-fold and achieved 60% coverage of male partners, the model estimated that 6987 incident HIV infections and 2603 deaths would be averted over a 10-year time horizon (Table 2). Discounted incremental costs of adding the HOPE intervention to standard of care (ART expansion) were \$14.3 million USD over 10 years with program model and \$9.9 million USD with the task-shifting model. The ICER for adding HOPE to standard of care was \$886 and \$615 USD per DALY averted for the program and task-shifting model, respectively. ICERs were similar with changing intervention coverage (40%-80%), although health benefits varied, as expected. At 40% coverage of male partners, HIV infections and deaths averted were reduced to 4659 and 1734, respectively, whereas 80% coverage was estimated to avert 9134 HIV infections and 3594 deaths. Lower intervention linkage to ART increased the ICER to \$1076 and \$730 for the program and task-shifting model, respectively, although ICERs were still below the threshold of Kenya's GDP per capita. Lower intervention linkage to ART at 60% coverage achieved greater reductions in HIV infections and deaths than higher intervention ART linkage at lower (40%) coverage. Similarly, higher HOPE coverage (80%) averted more HIV infections and deaths

			HOPE Intervention*								
	Current ART Scale-up	Base- Case	Lower Intervention Coverage (40%)	Higher Intervention Coverage (80%)	30% Lower Intervention Linkage to ART	30% Higher Intervention Linkage to ART	50% Lower ART Dropout	50% Higher ART Dropout	Lower Baseline ART Initiation	5% Increased Uptake of Option B+	
Total HIV infections	301,870										
Total deaths†	275,469										
Total DALYs	2,514,475										
HIV infections averted [‡]	—	6987	4659	9314	5480	8289	7293	6385	6398	8445	
Deaths averted‡	_	2603	1734	3474	2040	3090	2614	2539	2502	3304	
DALYs averted	_	16,192	10,788	21,601	12,651	19,270	16,282	15,769	15,268	20,419	
Incremental costs program model (millions)	—	14.3	9.6	19.1	13.6	15	14.5	14.1	13.9	15.3	
Incremental costs task- shifting model (millions)	_	9.9	6.6	13.3	9.2	10.6	10.0	9.7	9.5	10.9	
ICER program model (\$/DALY averted)	\$240	\$886	\$886	\$885	\$1076	\$778	\$888	\$892	\$911	\$749	
ICER task-shifting model (\$/DALY	\$240	\$615	\$614	\$614	\$730	\$549	\$616	\$616	\$621	\$533	

*Costs and infections are over 10-year time horizon. Incremental costs and DALYS associated with each strategy are discounted. HOPE intervention is added to current ART expansion. Percent coverage refers to coverage of male partners of pregnant women. Costs are in 2014 USD.

†All deaths, not only those related to HIV/AIDS.

‡Discounted health benefits.

compared with 60% coverage with higher ART linkage. The ICERs were robust to changes in ART dropout although intervention health benefits were lower at increased dropout rates. Because ART expansion in the next 10 years is uncertain, we assessed a scenario in which ART was rolled out at a lower rate (lower baseline ART initiation); this decreased intervention health benefits and resulted in higher ICERs-although they remained below the threshold of Kenya's GDP per capita. We also assessed a scenario in which undergoing couple HIV testing through the HOPE intervention increased women's initiation of Option B+. Results showed that increasing Option B+ initiation by just 5% yielded the highest benefits and lowest ICERs of all strategies at 60% coverage; ICERs were \$749 and \$533 for the program and task-shifting model, respectively. Overall, the HOPE intervention was projected to achieve 8% population coverage per year, as 14% of women are pregnant annually and 60% of couples would receive the HOPE intervention.

Figure 1 displays tornado diagrams of the sensitivity of ICERs for the program (1a) and task-shifting cost model (1b) to changes in the cost of HOPE, ART, and HIV/AIDS-related hospitalization. ICERs for both cost models were most sensitive to HOPE intervention costs, with the ICER exceeding the GDP for the program model at twice the intervention costs. ICERs were less sensitive to ART costs; higher ART costs resulted in less attractive (higher) ICERs and lower ART costs resulted in lower ICERs. The ICERs were least sensitive to hospitalization costs, which were inversely related to ICERs.

DISCUSSION

Women in Kenya attend antenatal care at rates over 90% and the majority are tested for HIV.¹⁹ However, they



*Orange line represents Kenya's GDP per capita, the threshold utilized for cost-effectiveness.

FIGURE 1. Tornado diagram of ICERs with varying costs. Orange line represents Kenya's GDP per capita, the threshold used for cost-effectiveness.

S178 | www.jaids.com

continue to experience disproportionately high HIV incidence, partially because of the lack of HIV testing and linkage for their male partners.⁹ Scaling up the HOPE intervention in the former Nyanza Province, Kenya can cost-effectively reduce HIV infections in pregnant women and their partners while averting morbidity and mortality associated with late initiation of ART in both sexes. Because men present to health care facilities at later stages of HIV and have lower ART linkage and poorer clinical outcomes, they are more likely to transmit HIV to their female partners.³⁶ Additionally, morbidity and mortality in men has a negative economic impact on women and their children. The HOPE intervention can identify HIV-positive men earlier in the course of their illness and link them to care before they infect their pregnant partners.

The projected health benefits of the HOPE intervention varied depending on the coverage achieved, the intervention's ability to link male partners to ART, and the baseline levels of ART initiation and dropout. However, the ICERs were robust to changes in these parameters and remained very costeffective (Table 2). ICERs were most sensitive to increases in intervention costs, with the program model exceeding the cost-effectiveness threshold of Kenya's GDP per capita at double intervention costs (Fig. 1). However, the detailed microcosting completed as part of the analysis increases our confidence in the estimated costs. Furthermore, the taskshifting model remained cost-effective despite increased intervention costs. In addition, if the HOPE intervention increases women's adherence to ART, as found in previous studies, the health benefits increase substantially and the ICERs become more attractive.¹⁶

Scaling up the HOPE intervention will likely be more affordable if implemented under a task-shifting model as staff salaries account for the majority of program costs. Training community health workers to conduct health interventions is an increasingly used strategy in SSA; community health workers can fill in service gaps caused by the shortage of health care professionals and deliver health care to rural areas more efficiently.³⁷ Integrating health interventions is another way to reduce program costs. For example, community health workers can deliver the HOPE intervention along with diabetes and hypertension screening. Integrating services can also reduce the stigma associated with an HIV testing intervention.³⁸ Additionally, a tiered program can be implemented in which women are given a few weeks to bring their partner into the clinic for testing. Those who do not present to the clinic can be followed up at home. Furthermore, a risk score could be developed to identify male partners who are less likely to come to the clinic so they can be traced at home. Such a score has been developed for HIV intimate partner notification in Malawi and has been shown to have good sensitivity and specificity.³⁹ Additionally, the HOPE intervention can be used to identify serodiscordant couples to target pre-exposure prophylaxis (PrEP), further integrating HIV services and decreasing costs. Finally, the HOPE intervention can be implemented within the context of home or mobile HTC for the general population. Although HOPE is cost-effective, it has a limited population-level impact on HIV burden as it is targeted to a specific portion of the population.

Copyright © 2016 Wolters Kluwer Health, Inc. All rights reserved.

Overall, it is projected to reach 8% population coverage annually and reduce approximately 2.5% of HIV infections in Nyanza over 10 years. Therefore, it should be combined with community-based HIV interventions that have been found to be cost-effective in other settings.³⁵

The strengths of our analysis include obtaining primary cost and effectiveness data from a randomized controlled trial in Nvanza, Kenva. Our results should be interpreted within the context of our limitations. We assume that male partners testing HIV-positive through the HOPE intervention have the same rate of linkage to ART as facility testing. Communitybased testing can result in lower linkage rates because it is conducted outside the health care system.⁴⁰ If implemented, the HOPE intervention should be monitored for linkage, and community health workers will likely need to conduct followup home visits or make phone calls to encourage reluctant partners to link to care. However, if the HOPE intervention increases adherence to ART and prevention of mother to child transmission, health benefits could be greater than projected. The effectiveness estimates were obtained from a randomized controlled trial that used highly trained and closely monitored health advisors. If the HOPE intervention is scaled up through a government program, effectiveness may decrease if counselors were less able to persuade reluctant male partners to undergo testing, or were less efficient in delivering the intervention. We explored a scenario of low intervention effectiveness (only 30% increase in linkage to ART) and HOPE remained cost-effective with the taskshifting scenario, although the intervention with full program costs exceeded Kenya's GDP per capita (Supplemental Digital Content, http://links.lww.com/QAI/A828.). As expected, the health benefits were considerably lower. This highlights the importance of follow-up to ensure linkage to care. Additionally, implementing standardized training and monitoring for community health workers would be essential in maintaining intervention fidelity. Currently, community health workers in Kenya are not trained to conduct HTC. Therefore, efficiency may be lower than expected during the first few years of the intervention as community health workers solidify their HIV testing skills. Indeed, the translation of a clinical trial to real-world implementation poses significant challenges, and the HOPE intervention would be most successful if it is integrated into other health interventions and uses community health workers instead of higher cadre staff.

In conclusion, we find the HOPE intervention to be a cost-effective method to reduce HIV disease burden in Nyanza, Kenya. Our results are similar to previous analyses that have found community-based HIV counseling and testing in SSA to be cost-effective.^{35,41} Although absolute costs will vary, our results are likely generalizable to other regions of SSA with a similar HIV epidemic profile.

REFERENCES

- UNAIDS. 2013 Report on the Global AIDS Epidemic. Available at: http:// www.unaids.org/en/resources/campaigns/globalreport2013/globalreport. Accessed May 22, 2015.
- Leroy V, Van de Perre P, Lepage P, et al. Seroincidence of HIV-1 infection in African women of reproductive age: a prospective cohort study in Kigali, Rwanda, 1988-1992. *AIDS*. 1994;8:983–986.

Copyright © 2016 Wolters Kluwer Health, Inc. All rights reserved.

- Taha TE, Dallabetta GA, Hoover DR, et al. Trends of HIV-1 and sexually transmitted diseases among pregnant and postpartum women in urban Malawi. *AIDS*. 1998;12:197–203.
- Humphrey JH, Hargrove JW, Malaba LC, et al. HIV incidence among post-partum women in Zimbabwe: risk factors and the effect of vitamin A supplementation. *AIDS*. 2006;20:1437–1446.
- Kinuthia J, Kiarie JN, Farquhar C, et al. Cofactors for HIV-1 incidence during pregnancy and postpartum period. *Curr HIV Res.* 2010;8:510–514.
- Gray RH, Li X, Kigozi G, et al. Increased risk of incident HIV during pregnancy in Rakai, Uganda: a prospective study. *Lancet*. 2005;366: 1182–1188.
- Calvert C, Ronsmans C. The contribution of HIV to pregnancy-related mortality: a systematic review and meta-analysis. *AIDS*. 2013;27:1631–1639.
- WHO. Global Update on the Health Sector Response to HIV. 2014. Available at: http://apps.who.int/iris/bitstream/10665/128494/1/9789241507585_ eng.pdf?ua=1. Accessed Ocotber 16, 2015.
- UNAIDS. Kenya AIDS Response Progress Report. 2014. Available at: http:// www.unaids.org/sites/default/files/country/documents/KEN_narrative_report_ 2014.pdf. Accessed January 20, 2016.
- Bellan SE, Fiorella KJ, Melesse DY, et al. Extra-couple HIV transmission in sub-Saharan Africa: a mathematical modelling study of survey data. *Lancet*. 2013;381:1561–1569.
- Kranzer K, Govindasamy D, Ford N, et al. Quantifying and addressing losses along the continuum of care for people living with HIV infection in sub-Saharan Africa: a systematic review. J Int AIDS Soc. 2012;15:17383.
- UNAIDS World AIDS Day Report 2012. Available at: http://www.unaids. org/sites/default/files/media_asset/JC2434_WorldAIDSday_results_en_1. pdf. Accessed May 25, 2015.
- Paulin HN, Blevins M, Koethe JR, et al. HIV testing service awareness and service uptake among female heads of household in rural Mozambique: results from a province-wide survey. *BMC Public Health*. 2015;15:1388.
- Suthar AB, Ford N, Bachanas PJ, et al. Towards universal voluntary HIV testing and counselling: a systematic review and meta-analysis of community-based approaches. *PLoS Med.* 2013;10:e1001496.
- Ebuy H, Yebyo H, Alemayehu M. Level of adherence and predictors of adherence to the Option B+ PMTCT programme in Tigray, northern Ethiopia. *Int J Infect Dis.* 2015;33:123–129.
- Becker S, Mlay R, Schwandt HM, et al. Comparing couples' and individual voluntary counseling and testing for HIV at antenatal clinics in Tanzania: a randomized trial. *AIDS Behav.* 2010;14:558–566.
- Rosenberg NE, Mtande TK, Saidi F, et al. Recruiting male partners for couple HIV testing and counselling in Malawi's option B+ programme: an unblinded randomised controlled trial. *Lancet HIV*. 2015;2:e483–e491.
- Cornell M, McIntyre J, Myer L. Men and antiretroviral therapy in Africa: our blind spot. *Trop Med Int Health.* 2011;16:828–829.
- Kenya AIDS Indicator Survey 2012. Available at: http://www.nacc.or.ke/ images/documents/KAIS-2012.pdf. Accessed January 1, 2016.
- Krakowiak D, Kinuthia J, Osoti AO, et al. Home-based HIV testing among pregnant couples increases partner testing and identification of serodiscordant partnerships. J Acquir Immune Defic Syndr. 2016;72 (suppl 2):S167–S173.
- Ying R, Sharma M, Heffron R, et al. Cost-effectiveness of pre-exposure prophylaxis targeted to high-risk serodiscordant couples as a bridge to sustained ART use in Kampala, Uganda. *J Int AIDS Soc.* 2015;18 (4 suppl 3):20013.
- Donnell D, Baeten JM, Kiarie J, et al. Heterosexual HIV-1 transmission after initiation of antiretroviral therapy: a prospective cohort analysis. *Lancet.* 2010;375:2092–2098.
- Salomon JA, Vos T, Hogan DR, et al. Common values in assessing health outcomes from disease and injury: disability weights measurement study for the Global Burden of Disease Study 2010. *Lancet.* 2012;380:2129–2143.
- 24. Mills EJ, Bakanda C, Birungi J, et al. Mortality by baseline CD4 cell count among HIV patients initiating antiretroviral therapy: evidence from a large cohort in Uganda. *AIDS*. 2011;25:851–855.
- Jahn A, Floyd S, Crampin AC, et al. Population-level effect of HIV on adult mortality and early evidence of reversal after introduction of antiretroviral therapy in Malawi. *Lancet.* 2008;371:1603–1611.
- MATLAB and Statistics Toolbox Release. Natick, MA: The MathWorks, Inc; 2015.
- World Bank Development Indicators: Antiretroviral Therapy Coverage (% of People Living with HIV). Available at: http://data.worldbank.org/indicator/ SH.HIV.ARTC.ZS?page=1. Accessed February 2, 2016.

- Drummond M, Drummond M. Methods for the Economic Evaluation of Health Care Programmes. 3rd ed. Oxford; NY: Oxford University Press; 2005.
- UNAIDS. Costing Guidelines for HIV Prevention Strategies. Geneva, Switzerland: UNAIDS; 2000. Available at: http://data.unaids.org/Publications/IRCpub05/jc412-costguidel_en.pdf. Accessed March 3, 2016.
- Obure CD, Vassall A, Michaels C, et al. Optimising the cost and delivery of HIV counselling and testing services in Kenya and Swaziland. Sex Transm Infect. 2012;88:498–503.
- Tagar E, Sundaram M, Condliffe K, et al. Multi-country analysis of treatment costs for HIV/AIDS (MATCH): facility-level ART unit cost analysis in Ethiopia, Malawi, Rwanda, South Africa and Zambia. *PloS One*. 2014;9:e108304.
- 32. Meyer-Rath G, Brennan AT, Fox MP, et al. Rates and cost of hospitalization before and after initiation of antiretroviral therapy in urban and rural settings in South Africa. J Acquir Immune Defic Syndr. 2013;62:322–328.
- Gold MR. Cost-effectiveness in Health and Medicine. New York, NY: Oxford University Press; 1996.
- Fasawe O, Avila C, Shaffer N, et al. Cost-effectiveness analysis of Option B+ for HIV prevention and treatment of mothers and children in Malawi. *PloS One.* 2013;8:e57778.

- 35. Smith JA, Sharma M, Levin C, et al. Cost-effectiveness of community-based strategies to strengthen the continuum of HIV care in rural South Africa: a health economic modelling analysis. *Lancet HIV*. 2015;2:e159–e168.
- Cornell M, Myer L. Men and mortality: sex inequality in health outcomes. AIDS. 2013;27:849–850.
- One million community health workers. *Technical Task Force Report*. Available at: http://frontlinehealthworkers.org/wp-content/uploads/2014/ 09/CHW-Report.pdf. Accessed January 24, 2016.
- Chamie G, Clark TD, Kabami J, et al. A hybrid mobile approach for population-wide HIV testing in rural east Africa: an observational study. *Lancet HIV*. 2016;3:e111–e119.
- Brown LB, Miller WC, Kamanga G, et al. Predicting partner HIV testing and counseling following a partner notification intervention. *AIDS Behav.* 2012;16:1148–1155.
- Sharma M, Ying R, Tarr G, et al. Systematic review and meta-analysis of community and facility-based HIV testing to address linkage to care gaps in sub-Saharan Africa. *Nature*. 2015;528:S77–S85.
- Bassett IV, Govindasamy D, Erlwanger AS, et al. Mobile HIV screening in Cape Town, South Africa: clinical impact, cost and cost-effectiveness. *PloS One.* 2014;9:e85197.